Author's response to reviews

Title: The subaortic tendon as a mimic of hypertrophic cardiomyopathy

Authors:

james j ker (jker@wol.co.za)

Version: 2 Date: 21 June 2009

Author's response to reviews:

Cover letter:

Dear Sir/Madam

1. I have expanded on the differential echo diagnosis in HCM.
2. Due to financial constraints there is no MRI or 3D images
3. I have uploaded more images

Thank you very much
Your`s sincerely